%\nocite{foley2021ad-software-engineering}
\nocite{sudermann2021mobile}
%\nocite{foley2021dindesign}
\nocite{des2021}
%\nocite{foley2021descomplexitykitchen}
%\nocite{foley2019complexitykitchen}
\nocite{icad2018book}
%\nocite{vossebeld2018customer}%01015
%\nocite{helgason2018tablet}%01015
\nocite{foley2017antenna}%01015
%%\nocite{foley2017faucet}%01009
%%\nocite{foley2017motorcyclelift}%01008
%%\nocite{foley2017adjustadesk}%01002

%\nocite{breznik2017collaborative}
%\nocite{foley2017msddontology}
%\nocite{petursson2017smaspring}

%%% Local Variables:
%%% mode: latex
%%% TeX-master: "foley-europass"
%%% End:
